Author's response to reviews

Title: Application of a plain abdominal radiograph transition zone (PARTZ) in Hirschsprung's disease

Authors:

 Akshay Pratap (akshaypratap2000@gmail.com)
 Devendra K Gupta (devendra6@hotmail.com)
 Awadhesh Tiwari (awadhesht03@yahoo.co.uk)
 Arvind K Sinha (arvindak1969@yahoo.com)
 Nisha Bhatta (nishakesharybhatta@yahoo.com)
 Chandra S Agrawal (drcsagrawal@yahoo.com)
 Shailesh Adhikary (shaileshadhikary@hotmail.com)
 Anand Kumar (anand_bhu@satyam.net.in)
 Sayendra N Singh (drsnsingh_gp@yahoo.com)

Version: 5 Date: 24 January 2007

Author's response to reviews:

Dear Sir,
Please find the revised version of the accepted manuscript with only changes done according to the journal policies.
Sincerely
Dr Akshay Pratap